Author's response to reviews

Title: Pulmonary artery intimal sarcoma a case report

Authors:

Hongqing HQ Long (longhongqing@163.com)
Qin Q QIN (luyin12340613@hotmail.com)
Conghua CH Xie (chxie-65@hotmail.com)

Version: 2 Date: 16 October 2007

Author's response to reviews: see over
this manuscript was revised by one of our professor, however, there is no big change. basically, the content is as same as before with little modification.